Author's response to reviews

Title: Spontaneous rupture of the long head of the biceps tendon in a hypothyroid patient: A case report

Authors:

Konstantinos Pantazis (pantazis.kon@gmail.com)
Dimitrios N Roupas (nikosroupas@hotmail.com)
Andreas Panagopoulos (andpan21@gmail.com)
Stauroula Theodoraki (roula-theodor@hotmail.com)
Aggeliki Tsintoni (agtsintoni@gmail.com)
Venetsana Kyriazopoulou (vkyriazopoulou@med.upatras.gr)

Version: 2 Date: 4 May 2015

Author's response to reviews:

After the initial assessment from your editorial board we made all the appropriate changes according the format of your journal

We are unable to find a similar case in the literature and probably this may the first report of this kind

It will significantly advance our understanding of hypothyroism in specific orthopaedic disorders. So both the orthopaedic surgeon and endocrinologist would be aware of this rare condition

It is an original case report of interest that have a broader clinical impact across more than one area of medicine